Joint Field Office Situation Report 
Human Services

Date: 10/9/05

1603:
	Registrations
	1,025,795

	Total Payments
	$2,154,253,344.14

	Designated Parishes 
	31


1607:

	Registrations
	296,239

	Total Payments
	$250,288,429.47

	Designated Parishes 
	21


	DRCs 

	Fixed
	30

	Mobile
	2


	Section Reporting: 
	Other Federal Agencies

	Group Leader Contact Info:
	Clara Feldberg

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603:  

US Rural Development

Talking Points regarding their loan and grant programs for those affected by Hurricane Katrina were distributed to Other Federal Agencies Coordinators and NPSC for dissemination.

Most of the Federal Agencies have employees on site at the DRCs:

Social Security Administration, Disaster Unemployment Assistance, Internal Revenue Service, USDA-Rural Development. Other Federal Agencies continue visiting DRCs periodically, leaving brochures and phone numbers for applicants looking for Information. 

Coordinating with Office of Victims of Crime offer through one of   it’s grantees to assist FEMA in death notifications efforts to victims of           Katrina.   

Offer has been declined because FEMA does not get involved in

Death notifications

1607:  US Rural Development

Talking Points regarding their loan and grant programs for those affected by Hurricane Rita were distributed to Other Federal Agencies Coordinators and NPSC for dissemination.

 Most of the Federal Agencies have employees on site at the DRCs:

Social Security Administration, Disaster Unemployment Assistance, Internal Revenue Service, US Small Business Administration, USDA-Rural Development. Other Federal Agencies continue visiting DRCs periodically, leaving brochures and phone numbers for applicants looking for Information. 

	
	B.
	Action Items for next Operational period
	1603:        Continue coordination with OFAs in a daily basis

1607:              US POSTAL SERVICE

Post Office trailers with post office boxes inside will be located in front of the cruise ship Sensation. Keys to the PO Boxes will be issue once the trailers arrive. Residents will need to show  photo identification and their names will be checked against occupancy lists that show cabin assignments.

	
	C.
	General Status of Operations
	1603:  on target

1607: on target

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	1
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:  n/a
1607: n/a


	Section Reporting: 
	HS IA Other Programs/Special Needs Group

	Group Leader Contact Info:
	Margaret Braun 225-346-4157; 703-547-7668

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603/1607: 

· Determined means to obtain a copy of the Coroner’s report of disaster related deaths, attempting to acquire copy. 

· Copy of specifications of ADA units on file at HQ. 

	
	B.
	Action Items for next Operational period
	1603/1607: 

· Continue to gather input from Group Leads to determine agenda for Special Needs Housing meeting on 10/23/05; time pending.  

	
	C.
	General Status of Operations
	1603/1607: Successes 

CR referral: 32 year old quadriplegic (evacuated from Kenner to Alexandria) with medical needs. Working through L.A.T.A.N. and Goodwill, managed to provide necessary medical commodities: specialized wheelchair, shower chair, and braces. Additionally, our referrals in concert with medical providers, have served as a secondary support network for the applicant. Applicant is now safely back in his dwelling and doing well.
PaRR referral: Kidney transplant patient, who lost a loved day of disaster, was evacuated to Kenner, and was residing with 12 other evacuees in very cramped living conditions, within approximately 72 hours applicant was housed in a travel trailer on her private site. 

CR referral: Applicant and her visually impaired husband were evacuated to a private shelter in New Roads, husband needed a wheel chair, which was obtained through L.A.T.A.N. Applicant mentioned that she hoped to relocate to Houston, TX to be near her son; referred applicant to the Facilitated Relocation Program; follow-up, NEMIS comment on 10/08/05: “USDA REFERRED TO NT (National Travel), APP AND HER HUSBAND WILL BE TRAVELING TO HOUSTON, TX."

1603:   
Special Needs (SN): Referrals Cumulative: 388; Pending: 25.
Crisis Counseling (CC):. 

Disaster Unemployment Assistance (DUA): Ongoing.

Disaster Legal Services (DLS: Ongoing.

Other Needs Assistance (ONA): No funerals have been paid. The Lump Sum Award is pending a final review by HQ’s legal counsel. 
1607:

Special Needs (SN): Referral Cumulative:10; Pending: 0.
Crisis Counseling (CC): Initiation of program pending.
Disaster Unemployment Assistance (DUA): Initial funding for hurricane Rita will is forthcoming. 
Disaster Legal Services (DLS): Initiation of program pending.
Other Needs Assistance (ONA): Ongoing.

	
	D.
	Staffing Needs
	

	
	
	
	(3) Pending completion of DAE hiring process.
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	9
	(3)
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603/1607: 


	Section Reporting: 
	Registration Sweep Teams

	Group Leader Contact Info:
	Pat Lord

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

· We have completed all of the major shelters for registration intake.

· We have transitioned some staff to the DH Transitional Team

1607:  
· We have completed over 1400 registrations.

	
	B.
	Action Items for next Operational period
	1603:    Complete special requests for registration intake. 

1607:     Determine what additional shelters we need to visit.

	
	C.
	General Status of Operations
	1603:

· 1SWEEP TEAM IS OUT ON THE FIELD TODAY.

· We have 13 staff members to respond to all requests

· The State is contacting all affected Parishes to see if they need a sweep team.

· We are finding most people are registered and have program questions.

1607:

· 2 SWEEP TEAMS ARE OUT IN THE FIELD TODAY. 

· Checking ARC list to see if any shelters need our help with registrations.

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	2 FEMA Staff / 18 contract staff 
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:
1607:


	Section Reporting: 
	ESF-6 Mass Care

	Group Leader Contact Info:
	Roger Ince; Desk: 225-334-7758; Cell: 916-541-0683

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603/1607:

1. ARF: Continuing to submit ARF to support kitchen supplies for ARC and TSA.

2. Miscellaneous: Staffing has been contacted and one person is arriving 10/10/05 (Howard Hinkson) to replace Baton Rouge staff. Staff for Lake Charles AFO (Steve Schroder) is deployed to Lake Charles. Staff for New Orleans AFO deployed to New Orleans (Constantin Cheremeteff).

	
	B.
	Action Items for next Operational period
	1603/1607: Working with ESF-1, ESF-5, PAO, ARC and housing to coordinate the effort of FEMA moving evacuees from shelters to trailers.

	
	C.
	General Status of Operations
	1603/1607: 

Operations are ongoing—shelters and feeding operations are normal, no problems.

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	4 at JFO; 1 in NOLA
	1 (replacement staff)
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603: N/A
1607:N/A


	Section Reporting:  
	VOLAG

	Group Leader Contact Info: 
	Harry Noftsker

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

1607:

	
	B.
	Action Items for next Operational period
	1603:

1607:

	
	C.
	General Status of Operations
	1603: 

GOODWILL:

Goodwill secured the donation of 23 FRIO packs to keep insulin or other injectable medications cool without refrigeration.  

Goodwill also secured the donation of three manual wheelchairs, andl will deliver them to the Office of Emergency Management for Plaquemines Parish on Saturday, October 8th.
Goodwill facilitated the readmit of a paraplegic evacuee back into the LSU Field House Special Needs Shelter. 

ISLAMIC RELIEF:

Distribution

Valerie’s Cajun Kitchen

· 10-7-2005

· Mac and cheese 16 boxes 30 lbs

· Ground beef sirloin 48 lbs

· Beef chunk tender steak 30 lbs

· Beef cubed steak 9.4 lbs

· Beef loin T-Bone steak 15.2 lbs

· Beef eye of round steak 13.4 lbs

· Beef chunk 7.1 lbs

· Biswich cookies 12 cases

· Bleach 30 bottles

· Cleaning kits 2 buckets

· Diapers size 4 11 cases

· Liquid detergent 9 cases

· Andy caps 1 box

· Tilex 1 case 

A Place of Refuge

· 10-7-2005

· Beef bottom round roast 15.6 lbs

· Boneless beef eye of round roast 16.9 lbs

· Ground beef round patty 12 lbs

· Beef loin porter house steak 11.1 lbs

· Ground beef round chili meat 12.1 lbs

· Beef sirloin lip steak 8.7 lbs

· Beef loin porter house steak 16.1 lbs

· Lean ground beef patty 12 lbs

· Tilex 2 cases

· Clear liquid detergent 2 cases

· Gain liquid detergent 2 cases

· Toothpaste 4 cases

UNITED SIKHS:

October 6, 2005

· Gave supplies including blankets to Ward 7 Shelter in Chauvin, and assessed their future needs, including those regarding after the shelter closes.  We also arranged dates for medical assistance.  

· Spoke to Ward 7 about their financial needs, reiterating the information we received from FEMA at previous meetings.

· Located shelters in Montegut, which will replace Ward 7 from 10/10/05.

· Confirmed arrangements for preparing and serving lunch at Allen Chapel.

· Helped Allen Chapel locate hotlines for donations, from where they could locate future supplies.

What We Have to Offer:

· A volunteer to work on any animal/pet projects
October 7and 8, 2005

· Received supplies of toiletries and personal hygiene products from Houston, Texas. 

· Prepared a lunch for the residents of Allen Chapel, which included rice, beans and potatoes and green beans.  This was kept vegetarian in keeping with Sikh tradition of a vegetarian free kitchen.

· Distributed supplies to Allen Chapel, including drinks, snacks, apple sauce and pillows.

· Continued to assess the needs of Sikh families in the area, and carry out our research in this area.

· Stock Check of supplies we still have left and those we still need

· Continue research into minorities as service-providers in emergency situations.

· Continue research on the effect of the hurricane on Sikh families in the region.

· Press and media work regarding our work in Louisiana, including a press release on the effects of the hurricane to Sikh families in the region.

· Updated our Shelter List, ensuring we specify which shelters feel they now have enough supplies, and which are looking for more.

DONATIONS:

While visiting sites in Houma Saturday we found a distribution site. Called Baby Central. They are distributing everything a child needs, the warehouse in New Iberia will be sending the 200 baby pack and go cribs along with Strollers, and baby swings. These are new items.

We will be receiving 40,000 blankets from the Macedonian Government in the next couple of weeks. 

ARC:

General Status of Red Cross Operations in Louisiana:

91 shelters this period 256 shelters to date

12,112 population (decrease of 2,258 since 10/06) 141,081 population to date

Meals: Period 189,750

            To date  7,864,643
 Snacks Period 274,754 to date 12,497,272

Community Services:

As of 10-09-2005 more than 21,453,000 have been served in the areas affected

by hurricanes Katrina and Rita.

There are 15 kitchens cooking in Louisiana, along with ongoing catering operations.

COS is supporting a kitchen in Algers in Orleans Parish in addition to supporting the distribution of water, food, clean-up supplies and baby supplies at several points of distribution (PODS) in the parish.

Individual Client Services: There are now 1.076,067 cases opened.

Families Re-Connecting

Calls to 1-877-LOVED1S total 193,422 with 1,204 calls received in the past 24 hours. 

AMURT:

AMURT/EL Information Referral: Two evacuees given housing information and referral to Baton Rouge DRC. One evacuee's information passed along to Red Cross regarding lost card. Two mental health professionals referred to Capital Area Human Services District. Shared information with Church of Scientology Volunteer Ministers in VOLAG about Vermillion Parish Volunteer Reception Center and NVOAD's Emotional And Spiritual Support Conference Call.

AMURT/EL Clean-Up Crew of 6 collaborated with Vermillion Volunteer Reception Center to gut out home of an older couple. Plan to continue with another home.

 AMURT/EL 150 Living/Welcome Kits boxes unloaded and distributed to Evacuees as they arrived at Groom Rd. Trailer site.

 AMURT/EL Emotional Support provided at Vermillion Shelter to evacuees and volunteers. 
 TSA:
Food Served

 

10/8/2005

Running Total

Prepared Meals (Hot or Cold)

19,719

1,063,569

Sandwiches

 

0

141,508

Cold Drinks (Soda/Juice/Water)

26,542

1,066,621

Snacks (Donuts, Cakes, Chips)

18,711

686,650

Other (Specify)

 

60

42,122



	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	9
	1 to check in on 10/11/05
	6

	2.
	CURRENT ISSUES OR PROBLEMS: 

1603:

1607:


	Section Reporting: 
	ESF-11

	Group Leader Contact Info:
	Doug MacAllister

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	

	
	B.
	Action Items for next Operational period
	

	
	C.
	General Status of Operations
	On 10/08/05, the Animal Plant Health Inspection Service (APHIS) opened Superior Canal where 150 head were stranded, and rescued 70 head by barge in Cameron & Calcasieu Parishes.  In Vermillion Parish, 80 head were rescued, and resources (panels and trailers) were being relocated to the central/western side of the Parish.  In Plaquemines Parish, hay delivery operations continued to cattle and horses in need, under the direction of Louisiana State University (LSU).


	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 

1603: 

1607: 


	Section Reporting: 
	Housing

	Group Leader Contact Info:
	Juan Gil/Maggie D.

	1.
	SITUATION:

	
	A.
	General Status of Operations
	· Placement  for families in Groom Road Site (East Baton Rouge Parish)  continues.  An additional 4  units  were leased in this morning , bring the total units leased to 264.  It is expected that a larger number of families will be available for leasing after church services today.

· Two(2) Transitional Teams are in Tangipahoa Parish identifying families  in shelters  for placemen in Ken Albin Mobile Home Site  and Dave’s Mobile Home Park, which now have units ready for occupancy.

· A third team is in Washington Parish shelters identifying families for placement into the Silver Creek Camppground travel trailers. 

	
	B.
	Major Priorities Accomplished
	RFO’S

· Ken Albin (Tangipahoa Parish)-10

· Melanie Rose (Lafayette Parish)-8

· Dave’s Mobile Home Park (Tangipahoe Parish)-13

· Treasure Wood (Jefferson Parish)-5

· Slidell I-59 (St Tammany Parish)-17

· Ledoux’s (Avoyelles Parish)-3

· Kemper Williams (St Mary Parish)-4

· Back to Back (Iberia Parish)-3

· Silver Creek (Washington Parish)-25

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff en route
	# of Staff still needed

	
	
	
	
	5
	0
	100

	2.
	CURRENT ISSUES OR PROBLEMS:  

· PPI is still in need of 20 additional Applicant Assistants.

· DHOPs is in need of 10 people to do Quality Assurance.  DHOPS will lose 23 Firemen on 10/7/05 that were requested back to their home destinations...

· The staffing situation for the Transitional Housing Team has become critical since the majority of the firefighters have left.  This section needs 75 Applicant Services Specialists as soon as possible.  They will be used in mass shelters to identify families for relocation and provide information on how the relocation will be done. They are also used at the group sites to do data intake on the families as they leased and at the JFO to do the advance communications with Managers of the shelters the teams will be visiting next.

· The Applicant Services section is also getting overloaded with congressional and other inquiries.  Staff has been requested, but is pending due to lack of space.


	Section Reporting:  
	U.S. SBA

	Group Leader Contact Info:
	Carl N. Gaspari

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: Customer Service Representatives are assigned to all DRCs to answer questions and assist individuals and business owners complete their SBA Application.  

A Personnelist is now on site to begin hiring local staff.

Identified location for a Business Recovery Center in NO

1607:

	
	B.
	Action Items for next Operational period
	1603: 

Identify available space for SBA staff at APOs and New Orleans.
1607:

	
	C.
	General Status of Operations
	1603: 

SBA continues to search out conventional lodging for members or our staff. 

Local hiring of CSRs and administrative staff is ongoing. 

Set up the Business Recovery Center in New Orleans.
Continue training of local hires in Baton Rouge.

Identified location for a Business Recovery Center in NO.
1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	LA 220  TX 76
	5
	228

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:  Locating lodging to support staff in close proximity to the DRCs and JFO.  

1607:


	Section Reporting: 
	Barbara Copeland/Kent Sheets

	Group Leader Contact Info:
	Inspection Service Coordinators

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

There are 1216 inspectors performing inspections. 

1607: 
There are 279 inspectors performing inspections.

	
	B.
	Action Items for next Operational period
	1603:
1607:

	
	C.
	General Status of Operations
	1603:

1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	2
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:
1607:


	Section Reporting: 
	NPSC Liaison

	Group Leader Contact Info: 
	Pamela Glasschroeder 703-785-2661

	1.
	SITUATION:

	
	A.
	General Status of Operations
	1603/1607: 

· Assisting with special needs, congressional and EOR cases

· Compiling reports and identifying trends 

· Relating information between the NPSCs and JFO

· Providing DRC support 

	
	B.
	Major Priorities Accomplished
	1603/1607

· Looked up applicants to see if they have registered with FEMA, for the MH/TT program

· Working on special needs cases and escalated cases.

· Answering program questions for varies entities within the JFO

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	0
	1

	2.
	CURRENT ISSUES OR PROBLEMS:  


	Section Reporting: 
	DRC OPS

	Group Leader and Contact Info:
	Don Baggett

	1.
	SITUATION:

	
	A.
	General Status of Operations
	DRC at Belle Chasse will be open daily from 8am – 6pm.  All other DRC’s will be open daily from 9am – 7pm.

 Fixed DRC sites now in 30 locations:

Caddo(Shreveport)

Ouacita(Monroe)

Rapides(Alexandria)

Lafayette(Lafayette)

Iberville(Plaquemine)

Plaquemines (Belle Chase)

Webster (Minden)

E. Baton Rouge (Baton Rouge)

Washington(Bogalusa)

Avoyelles(Marksville)

Washington (Franklinton)

St Tammany(Slidell)

Orleans(Algiers)

St James (Vacherie)

LaFourche(Thibodeaux)

St Charles (Boutte)

St John (LaPlace)

Ascension(Gonzales)

Jefferson(Gretna)

Iberia(New Iberia)

LaFourche(Raceland)

Tangipahoa(Amite)

Livingston (Denham Springs)

Terrebonne(Houma)

Point Coupee (New Roads)

St Tammany(Covington)

Vermillion(Abbeville)

Jefferson Davis(Jennings)  

W. Baton Rouge(Port Allen)  

 St Helena(Greensburg)  

MDRC’s:

Cruise Ship at Port of New Orleans

Washington Parish

We have also secured leases on properties at the following locations and will announce openings ASAP(In order of priority):

Tentatively Monday 10/10/05:

Orleans(New Orleans#1)

Tentatively Tuesday 10/11/05:

Cameron(Grand Lake)

Calcasieu (Lake Charles) 

Tentatively Thursday 10/13/05:

St Bernard(Chalmette)

Concordia(Vidalia) ASAP

Beauregard(DeRidder) ASAP

Lincoln(Ruston) ASAP

Natchitoches(Natchitoches) ASAP

Acadia(Rayne) ASAP

Morehouse(Bastrop) ASAP

St Martin(Breaux Bridge) ASAP

St Landry(Opelousas) ASAP

Red River(Coushatta) ASAP

Franklin(Winnsboro) ASAP

Caldwell(Columbia) ASAP

Tensas(St Joseph) ASAP

Jackson(Hodge) ASAP

Vernon(Leesville) ASAP

E Feliciana(Clinton) ASAP

Winn(Winnfield) ASAP
#1 Priority: To secure inspected, approved, and leased sites at the following Parishes:  

Orleans (New Orleans#2)

Orleans (New Orleans#3)

Sabine (Many)

LaSalle(Jena)

Evangeline(Ville Platte)

Richland(Rayville)

More locations possibly to follow.

	
	B.
	Major Priorities Accomplished
	

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff en route
 
	# of Staff still needed

	
	
	
	
	383
	225
	400

	2.
	CURRENT ISSUES OR PROBLEMS:   

Approximately 50% of the OFA and Firefighter personnel currently deployed to DRC’s will be reaching the end of their tour of duty by 10/07/05 and will have to be replaced.  We are using various resources to address this problem, including additional hiring under the IA TAC, and working with Admin to obtain Local Hires.

DFTO & IA Trainers are now on-site at the JFO to train all DRC personnel.
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1

